
SOAS ISFS Status for June 12
SOAS ISFS Status, June 12, 2013

Summary: 32m Licor out, w/ replacement in base trailer being configured.  20m Licor is out.  Not reporting at all.  4.4cm soil temp is still flaky.  8m sonic 
was flaky last night.

Sonics: ok, w/ 8m was flaky last night but good now.

TRH, P:  ok

Wind speed & direction: ok

Soil, Rainr: Tsoil.4.4cm bad, may add additional Tsoil to replace this unit (an Oncley install)

Vmote.(rad,a,b):  all approx. 13.0V

Radiation, Wetness: ok

h2o, co2: 32m Licor gone.  20m Licor sick.  Kryptons both around 0.2V.

System (GPS, Vdsm): ok 
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